Author's response to reviews

Title: The importance of comorbidity in analysing patient costs in Swedish primary care

Authors:

Sven G Engstrom (sven.engstrom@lio.se)
Lennart Carlsson (lennart.carlsson@chello.se)
Carl-Johan Ostgren (carl.johan.ostgren@lio.se)
Gunnar H Nilsson (gunnar.nilsson@nlpo.sll.se)
Lars A Borgquist (larbo@ihs.liu.se)

Version: 4 Date: 4 February 2006

Author's response to reviews:

My manuscript has been revised as requested:
1) key words have been removed
2) Table 2 has been changed in format from landscape to portrait
No other changes. No upload but for the main manuscript.
Regards,
Lennart